Author's response to reviews

Title: Isolated radial head dislocation, a rare and easily missed injury in the presence of major distracting injuries: a case report

Authors:
Ulfin Rethnam (ulfinr@yahoo.com)
Rajam S U Yesupalan (ajeesh2000@yahoo.co.uk)
Salah S Bastawrous (Salah.bastawrous@cd-tr.wales.nhs.uk)

Version: 5 Date: 25 June 2007

Author's response to reviews:

To
The Editor
Journal of Medical Case Reports.
Thank you for accepting our manuscript for publication in your esteemed journal. The manuscript has been formatted as suggested by you. The changes to the manuscript is mentioned below:
1. The title has been changed to "Isolated radial head dislocation, a rare and easily missed injury in the presence of major distracting injuries: a case report"
2. The title page is aligned to the left
3. The keywords have been removed from the manuscript
4. The abstract has been moved to the page after the title page
5. The subtitle "Introduction" has been changed to "Background"